



Cogent Business & Management

ISSN: (Print) (Online) Journal homepage: www.tandfonline.com/journals/oabm20

Long-term effect of the form of payment in Spanish mergers and acquisitions: an event study

Jesús Reglero, Teresa Corzo Santamaría & Rocío Saénz-Diez

To cite this article: Jesús Reglero, Teresa Corzo Santamaría & Rocío Saénz-Diez (2024) Longterm effect of the form of payment in Spanish mergers and acquisitions: an event study, Cogent Business & Management, 11:1, 2374878, DOI: <u>10.1080/23311975.2024.2374878</u>

To link to this article: <u>https://doi.org/10.1080/23311975.2024.2374878</u>

© 2024 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group



0

View supplementary material 🖸

•

Published online: 22 Jul 2024.

		•
--	--	---

Submit your article to this journal \square





View related articles 🖸



View Crossmark data 🗹

BANKING & FINANCE | RESEARCH ARTICLE

OPEN ACCESS OPEN ACCESS

c*gent

Long-term effect of the form of payment in Spanish mergers and acquisitions: an event study

Jesús Reglero^{a,b} (b), Teresa Corzo Santamaría^b (b) and Rocío Saénz-Diez^b (b)

^aRS Corporate Finance, Lausanne, Switzerland; ^bICADE School of Economics and Business Administration - Universidad Pontificia Comillas, Madrid, Spain

ABSTRACT

This paper explores the methods of payments for companies engaged in mergers or acquisitions, focusing on the Spanish market from 2007 to 2019, specifically on transformative transactions. We use a novel approach to build a sample of homogeneous corporate transactions that allows us to measure the performance for the acquirer one year after the announcement. A descriptive analysis, considering seventeen criteria, and an event study performed using the market model, CAPM and the Fama-French and Carhart four factor model, evaluate the acquirer's performance post-announcement. Statistically significative results indicate that using shares over cash is preferable for acquirers to achieve sustained success. These findings reveal that, despite theoretical advantages of cash, shares as method of payment may lead to better stock market performance, limited financial profile deterioration, and more favorable long-term prospects for shareholders.

ARTICLE HISTORY

Received 13 December 2023 Revised 26 May 2024 Accepted 13 June 2024

KEYWORDS

Mergers and acquisitions; method of payment; event study; transformative transaction; Spain

REVIEWING EDITOR

David McMillan, University of Stirling, United Kingdom of Great Britain and Northern Ireland

SUBJECTS

Finance; Economics; Corporate Finance; **Investment & Securities**

JEL CODES G1; G3

1. Introduction

Corporate growth and value creation are fundamental objectives for management teams, as emphasized by Friedman (1970), who underscores the responsibility of achieving maximum benefits for shareholders. To create value, companies must demonstrate that their decisions generate returns exceeding costs, forming a virtuous circle (Bughin & Copeland, 1997). Mergers and acquisitions (M&A), both vertical and horizontal, serve as instruments for managers to pursue these objectives (Pardo-Garcia & Sempere-Monerris, 2015), and although originated in the USA at the turn of the 20th century, they are currently considered a global phenomenon based on a variety of motives and success factors as highlighted by Calipha et al. (2010).

These corporate transactions have been considered contagious (Öberg & Holtström, 2006), and it is possible to identify periods of increased corporate activity in what are known as merger waves shaped by economic, regulatory, and technological changes, along with central bank policies (Chernenko et al., 2021; Harford, 2005). Mergers and acquisitions were surging in recent years, and even after the pandemic disruption, the post 2020 rebound was strong, reaching historical volumes of \$4.3 tn and 41,000 deals in 2021. Since then, 2022 slowed down to \$2.7 tn and 37,000 deals due to factors such as rising interest rates, geopolitical uncertainty, and global conflicts. Although this decline continued during the first half of 2023, during the last quarter of said year it looks like M&A activity was picking up and may be gathering momentum.¹

Despite the long history and established track record of M&A, there seems to be a popular view that this activity is unprofitable and many deals fail (Bruner, 2002; Grubb & Lamb, 2001; Straub 2007). The outcome of the transactions is affected by key determinants such as CEO overconfidence,

CONTACT Jesús Reglero 🖾 jesus.reglero@resacorporatefinance.com, jreglero@comillas.edu 🖃 RS Corporate Finance, Lausanne, Switzerland © 2024 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. The terms on which this article has been published allow the posting of the Accepted Manuscript in a repository by the author(s) or with their consent.

acquirer-target relatedness and complementarity or shareholder intervention (Renneboog & Vansteenkiste, 2019). Additionally, the method of payment plays a key role in the probability of failure together with target status, diversification, and acquirers prior experience (Tanna et al., 2020). However, the research on M&A success focuses on firms' performance around and after a takeover using a wide variety of performance measures and heterogeneous samples, which complicates drawing accurate and unambiguous conclusions. Thus, according to Renneboog and Vansteenkiste (2019) 'given the small number of M&A research areas that consistently predict deal success or failure, it remains essential to investigate both short- and long-run return measures when examining post-deal performance. Moreover, the lack of long-run evidence for many areas of the M&A literature provides substantial scope for further research.'

This paper presents an empirical study on how payment methods for M&A deals affect the acquirer's long term performance (one year after the announcement) in a sample of Spanish companies. Most of the existing research on this field comprises US or UK companies, although some works have studied other geographies like Canada (Eckbo et al., 1990 and Masse et al., 1990), Japan (Higgins, 2013), Australia (Diepold et al, 2008), Greece (Liargovas & Repousis, 2011), or The Netherlands (Swieringa & Schauten, 2007) among others. Alexandridis et al. (2010) carry out an empirical analysis with a global M&A sample and conclude that the acquiring firm's performance vary across countries depending on the competition of takeover markets. Similarly, Moschieri et al. (2014), Faccio and Lang (2002) or Feito-Ruiz et al. (2014) study European M&A activity analyzing cross-border activity, showing that M&A deals appear to be affected by country factors, like its legal and institutional environment. Therefore, multi-country studies need to be complemented with single country studies. In Spain, previous studies are limited and primarily concentrate on short-term effects, such as abnormal returns on the deal announcement date (see Farinós Viñas et al, 2017; Fernández & Gómez-Ansón, 1999; García & Ferrando, 1992; Miguel et al, 2003). Therefore, our research enhances the existing body of country-specific literature by focusing on Spanish transactions.

The research is based on the construction of a novel database of homogeneous transactions in the Spanish market that could properly assess long-term performance for the acquirer depending on the payment method. With this purpose in mind, and in order to keep a homogeneous and comparable set, it was necessary to find a reliable criterion to include or exclude transactions in the final sample, which finally included 39 transactions (all of them listed, with a Spanish component, either target or acquirer, and that meet the restrictive criteria to be useful for a long-term analysis). After a quantitative and qualitative description summary of the database, we carry out an event study measuring the impact on the acquirer on absolute terms, versus the reference indexes, and analyzing the abnormal returns as per the CAPM model (Sharpe, 1964) and the Fama and French (1993) model enhanced by Carhart (1997) four factor model (for convenience along the paper we will refer to this model as the Fama-French-Carhart). We confirm the statistical robustness of the research.

Findings show that methods of payment do have an effect in the acquirer's long term performance, showing higher abnormal returns on stock deals than in cash deals.

M&A research on methods of payment has always attracted a lot of attention, and there is extensive literature on the subject, as explained in the next section. The studies often focus on empirical tests, due to the practical and always evolving nature of this field and to the interest of this topic not only for academics but also for practitioners. Our paper is ingrained in these type of empirical studies on the effect of M&A consideration, showing the following contributions: (i) while most works study the short-term reaction immediately after the deal, we analyze the long-term effect of the transaction on the acquirer's performance, (ii) we improve over previous works that also covered long-term effects, since the contamination effect of other events on the acquirer is mitigated by building a homogeneous and comparable database of M&A with an innovative and valuable approach, (iii) our sample of Spanish companies adds to the body of works that focuses on local geographies, with a similar legal framework, enriching the international analysis of such a global phenomenon as M&A transactions.

The subsequent sections follow a structured approach: Section 2 provides a relevant yet non-exhaustive literature review, while Section 3 details the selected sample and applied methodology. Section 4 summarizes analyses, including sample analysis, event study, and statistical significance tests. Finally, Section 5 concludes the paper, highlighting avenues for future research.

2. Literature review

There are various approaches that have analyzed the method of payment of corporate transactions, from theoretical works (Eckbo et al., 1990; Fishman, 1989 or Hansen, 1987) to several empirical papers that focus on the choice of stock versus cash (Boone et al. 2014; Carleton et al., 1983; Eckbo & Langohr, 1989; Faccio & Masulis, 2005; Franks et al., 1991; Huang & Walkling, 1987; Martin, 1996, among many others). Works like Bruner (2002) or Sankar and Leepsa (2018) summarize different studies identifying a variety of factors that affect the means of payment in M&A deal.

The presence of information asymmetry causing a certain choice between cash and stock is behind many of the works, like Eckbo et al. (1990); Fishman (1989); Hansen (1987); Ismail and Krause (2010), Travlos (1987) and Yook et al. (1999), who mention the need to mitigate information asymmetry between the buyer and the target company by using the buyer's stock as the payment method, while cash payments are more common when there is greater clarity of information. On the same line, Shleifer and Vishny (2003) assume that markets are not rational, but acquirers' managers take advantage of paying with stock when they know their stock have high valuations, exploiting the false belief of the target company's shareholders, who may think that the offer is better than it is (Rhodes-Kropf & Viswanathan, 2004; Savor & Lu, 2009). Several studies show that stock acquisitions are more common when valuations are high, associating the payment method with the buyer's valuation (see Aguirreamalloa, 2013; Brown & Ryngaert, 1991; Dong et al, 2006; Farinós Viñas et al, 2017).

The literature on payment choice in M&A also rests on tax considerations, based on the different taxation of stock and cash payments (Auerbach & Reishus, 1988; Ayers et al., 2003, 2004; Houston & Ryngaert, 1997). Since paying with stock means there is no immediate tax effect for target shareholders, who defer tax payments until they eventually sell the new shares, some believe that cash offers would have a higher premium vis a vis stock offers to compensate the capital gain tax. Also, stock offers become more appealing when taxes are high. Several empirical works (e.g. Harris et al., 1987; Huang & Walkling, 1987; Wansley et al., 1983) support the role of tax when choosing the deal's payment form.

Continuing with possible approaches, Bharadwaj and Shivdasani (2003); Dong et al. (2006); Faccio & Masulis, 2005; Harford et al. (2009) and Martin (1996), show that the buyer's indebtedness is another variable to consider. The form of payment will therefore vary depending on the investment opportunities available (Martin, 1996; Swieringa & Schauten, 2007; Zhang et al., 2003) and the company's target leverage ratios (Harford et al. 2009; Uysal, 2011). Aguirreamalloa (2013) mentions payment methods are related to other variables according to the literature, such as the buyer's shareholding structure, since stock payments imply a potential dilution and therefore a change in the control *status quo* (Amihud et al., 1990; Blackburn et al., 1997; Eckbo et al, 1990; Faccio & Masulis, 2005; Harris & Raviv, 1988; Stulz, 1988; Yook et al, 1999) or the willingness to provoke transactions with a higher level of competition from other bidders (Fishman, 1989). Other factors studied are the preferences of target shareholders (Burch et al, 2012; Gaspar et al, 2005), the negotiating power of the target to accept stock based on the distance between the share price and the investor's reference point (Chira & Madura, 2018), the fact that the bid is solicited or unsolicited (Dong et al, 2006; Huang & Walkling, 1987; Schwert, 2000) or the relative size of both companies, the target and the buyer (Draper & Paudyal, 2006; Grullon et al., 1997; Swieringa & Schauten, 2007).

Based on the reviewed articles, there are several challenges that academia faces when creating knowledge in this area of corporate finance:

There is some debate on whether it is preferable to use short-term (e.g. Diepold et al., 2008) or long-term windows (e.g. Barber & Lyon, 1997) in event studies, since the evidence is unclear. On one hand, Healy et al. (1992), found a positive relation between short window abnormal stock returns and long-term post-acquisition increases in operating cash flows. On the other hand, as Zollo and Meier (2008) remind us, the use of long-term windows is conceptually preferred, given the evidence on market imperfection at the time of the announcement of complex events such as acquisitions. Harrison et al. (2005) demonstrated that while the market's immediate response to the announcement is positive, the long-term reaction is significantly negative. Furthermore, several control variables exhibit significant long-term effects, suggesting that the market fails to accurately foresee the economic consequences of the announced acquisitions, leading to systematic over- or under-reactions.

Most works analyze the choice of payment method for M&A transactions looking at its short term effect, that is, the impact on the acquirer's stock price around or on the same day of the announcement: Asquith et al. (1983); Boone et al. (2014); Brown and Ryngaert (1991); Eckbo et al. (1990); Franks and Harris (1989); Huang and Walkling (1987); Ismail and Krause (2010); Peterson and Peterson (1991); Savor and Lu (2009); Travlos (1987); Uysal (2011); Wansley et al. (1983); Yook et al. (1999). Some studies do analyze the longer-term impact, like Agrawal et al (1992) or Loughran and Vijh (1997), who consider a five-year post-merger period; Sullivan et al. (1994); who look at the period until the merger is finalized, same as Asquith (1983), who considers this period to be 240 trading days after the bid. Other studies look at a three-year period (Mitchell & Stafford, 2000; Rau & Vermaelen, 1998), or 30/36 months (Chakrabarti et al, 2009; Zollo & Meier, 2008). It is clear that long-term returns need to be controlled by other factors and its proper calculation is challenging besides market reaction considerations, companies undertake numerous actions over the long run that may affect the stock price.

As for empirical studies, one interesting approach is to study a case in-depth and extract the most relevant conclusions, trying to find valuable insights for the topic (e.g. Bruner, 1999; Bruner, 2015; Gielens et al., 2008; Kaplan 1989; Lys & Vincent, 1995). However, the most common approach is to build a sample of transactions in a specific period and in a specific region on which to test hypotheses. (Boone et al., 2014; Burch et al, 2012; Emery & Switzer, 1999; Feito-Ruiz et al., 2014; Harris et al., 1987; Heron & Lie, 2002; Higgins, 2013; Huang et al., 2012; Huang & Walkling, 1987; Ingham et al., 1992; Ismail & Krause, 2010; Karampatsas et al, 2014; Martin 1996; Masse et al, 1990; Pinkowitz et al, 2013; Ruback, 1983; Sullivan et al, 1994; Travlos, 1987; Uysal, 2011; Vermaelen & Xu, 2014; Wansley et al, 1983; Yook, 2003; Yook et al, 1999). These samples are usually large and combine different transactions, so they are not homogeneous, a fact which makes more difficult to interpret results and infer general insights. As Renneboog and Vansteenkiste (2019) remind us, 'wide variety of performance measures and heterogeneity in sample sizes complicates the drawing of accurate and unambiguous conclusions.'

Regarding the geographic scope, as previously mentioned, most existing research has focused on US data (Burch et al., 2012; Emery & Switzer, 1999; Ismail & Krause, 2010; Pinkowitz et al., 2013; Sullivan et al., 1994; Yook, 2003), or the UK (Franks & Harris 1989) although global studies are also available (Alexandridis et al., 2010; Faccio & Lang, 2002; Feito-Ruiz et al., 2014; Moschieri et al., 2014), and works with local samples (Eckbo et al., 1990; Farinós Viñas et al, 2017; Higgins, 2013; Masse et al., 1990; Swieringa & Schauten, 2007) are gaining popularity based on an interest to disentangle the effect of specific country factors in M&A deals. Spanish M&A activity ranks consistently in the top-five position in Europe after the United Kingdom, France, Germany, and the Netherlands as per Mergermarket annual reports of the last years, and its specific country idiosyncrasies, its links to Latin America, where many companies have sought expansion through inorganic growth and its legal and institutional framework make it a relevant market deserving attention for a corporate transaction study.

3. Methodology

3.1. Sample construction

To tackle the previously mentioned challenges, a novel approach to sample construction was developed, with specially selected deals that followed certain features and represent what we have called a 'Transformative Transaction'. A Transformative Transaction is defined as a corporate transaction in which, during the period from its announcement to one year later, the acquirer does not undertake any more relevant internal or external actions in the company other than (i) integrating the conducted transaction and (ii) investing all its time and resources into the organic business without any other significant changes. Often, a Transformative Transaction would represent a large deal, in terms of the relative size of target vs. buyer, that being one of the reasons why the acquirer spends the following year integrating the newly acquired company.

Focusing on Transformative Transactions will avoid most contaminating events that could impact the company performance, following Golubov et al. (2015) who stress the importance of accurately isolating

the takeover effect from other factors affecting the firm when measuring long-run returns. The resulting sample, built only with Transformative Transactions, will be more homogeneous and should led to more robust results when measuring long term performance.

A Transformative Transaction can be done by a serial on non-serial (single) acquirer. Serial acquirers are companies that have carried out a certain number of transactions during an analyzed timeframe (Billett & Qian, 2008; Fuller et al., 2002; Karolyi et al., 2015; Kengelbach et al., 2012). According to Renneboog and Vansteenkiste (2019), serial bidders are quite a common phenomenon: they were responsible for over 25% of M&A activity in the 1980s and 1990s (Klasa & Stegemoller, 2007), 75% of listed US firms were frequently engaged in corporate transactions during the 1990s and 2000s (Netter et al., 2011) and they performed 32% of public deals and 31% of private deals in the 2010–2015 period (Alexandridis et al., 2017). Karolyi et al. (2015) estimate that one in five public bidders is a serial buyer.

For our research, we include both serial and non-serial buyers in our sample, focusing on the specific Transformative Transactions per buyer.

The following Figure 1 provides a simplified outline of the definition of a Transformative Transaction.





Our sample of transactions is drawn from the CapCorpData dataset, provided by Capital&Corporate – Grupo Ifaes, a private Spanish data provider for Private Equity and M&A transactions. Looking at the period from 2007 to 2019, we initially obtained 2,325 deals in which at least one of the companies involved had to be Spanish. Transactions where the acquirer was not a listed company in an official stock exchange were excluded from this study, as were state owned companies. Also, only deals that could be categorized as a merger or an acquisition were included, excluding IPOs or block trades. After this first filter, we were left with 60 transactions.

The second filter used to confirm if the transaction is transformative was to analyze and identify, for each potentially qualifying transaction, all relevant events registered in the stock market supervisor where they buyer shares are traded, during the period from the announcement of the transaction to one year afterwards. For Spanish publicly traded companies, we checked the information provided by the Spanish market regulator, the Comisión Nacional del Mercado de Valores (CNMV), and for non-Spanish traded companies that had acquired a Spanish target, the relevant events disclosed in the corresponding market supervisor. This filter aimed to verify if the proposed definition for Transformational Transaction could be met. Our categorization produced a final sample with 39 Transformative Transactions with Spanish elements (buyer, target or both) for the period 2007–2019 that conform a homogeneous set for the purpose of our research. Although the size of the sample is reduced compared with related works about methods of payment in M&A, this is due to the restrictive requirement of no contaminating events, which entails a better comparability of the observations for the purposes of our analysis. Also, smaller samples are often used for country specific M&A research (e.g. Diepold et al, 2008 use a sample of 50 Australian deals), or for M&A event studies (Dilshad, 2013 carry out an event study with 18 transactions).

This process is summarized in Figure 2.



Figure 2. Sample construction methodology.

A summary of the whole sample is included in Table A1, and all its detailed data is available at Harvard Dataverse ('Spanish dataset or mergers and acquisitions 2007–2019', https://doi.org/10.7910/ DVN/2WK6ON). Most acquirers in the sample are companies that experienced only one Transformative Transaction within the timeframe considered, but a few, such as Iberdrola, BBVA, Santander, and Grifols, completed more than one transaction.

3.2. Sample descriptive analysis

First, we describe our database of 39 Transformative Transactions, analyzing seventeen quantitative and qualitative indicators related to the acquirer. We compare overall outcomes and distinguish between transactions that used cash and those that used stock. The detailed indicators for the acquirer, include:

- 1. Absolute stock market performance measured as the change in the stock price.
- 2. Dividend yield.
- 3. Total shareholder return (the sum of stock market performance and dividend yield).
- 4. Relative stock market performance compared to the reference market index.
- 5. Valuation multiple defined as enterprise value to EBITDA.
- 6. Valuation multiple defined as market capitalization to net profit.
- 7. Valuation multiple defined as market capitalization to book value.
- 8. The profitability ratio calculated as net profit over equity.
- 9. The profit margin ratio calculated as net profit over revenue.
- 10. Asset turnover ratio calculated as revenue over total assets.
- 11. Leverage ratio calculated as total assets over equity.
- 12. Does the acquirer need to reduce dividends?
- 13. Does the acquirer need to issue additional capital?
- 14. Does the acquirer need to sell assets to reduce debt?
- 15. Does the acquirer need to change the Chairman or CEO?
- 16. Does the acquirer experience negative changes in the rating?
- 17. Does the acquirer need to undergo a restructuring process?

3.3. Event study

After the descriptive analysis, an event study was conducted, defining the event day as the announcement day of the Transformative Transaction, trying to determine whether the transaction itself had an impact on the long-term performance of the acquirer, and if this impact differs between transactions paid with cash and those paid with stock. The return of the acquirer has been measured on absolute terms, assuming a one year buy and hold strategy (BHAR) (see Mitchell & Stafford, 2000 for a revision of flaws when applying BHAR), and calculating the abnormal return of the acquirer taking as a benchmark the performance of the market index, the cost of equity as per the Capital Asset Pricing Model (CAPM) and the cost of equity as per Fama-French-Carhart four factor model. As Kothari and Warner (2007) point out 'while the exact definition of long horizon is arbitrary, it generally applies to event windows of 1 year or more', as is the case of our study. Statistical inference is applied to confirm the significance of the results.

Event studies have a long academic history, (Ashley, 1962; Barker, 1956; Dolley, 1933; Myers & Bakay, 1948). However, two articles from the late 1960s were fundamental to this research field: Ball and Brown (1968) and Fama et al (1969) introduced the methodology for conducting these studies. Ball and Brown applied it to firms' financial results, while Fama studied the effects of splits, isolating the effect from simultaneous dividend increase announcements.

Since then, event studies have become a widely used methodology to examine the impact of firm-specific and economy wide events on the value of a firm, with applications in the field of accounting and finance, like earnings announcements, debt issuances, capital increases, and macroeconomic events, among others. Reviewing the immense event studies literature exceeds the scope of this paper although many examples could be mentioned in the M&A field (Antoniadis et al., 2014; McAfee & Williams, 1988; Shah & Arora, 2014), where event studies have been used numerous times: analyzing transactions in certain industries, like the banking sector (Liargovas & Repousis, 2011; Dilshad, 2013), internet (Uhlenbruck et al., 2006; Zaheer et al., 2010) or renewable energy (Yoo et al., 2013); measuring the profitability of M&A deals (Capron & Pistre, 2002; Duso et al., 2010; Diepold et al, 2008; Elad & Bongbee, 2016; Oler et al., 2008), or the effect prior to the announcement date (Rani et al., 2015). Also, event studies have been used for studies with a country focus (Akben-Selcuk, 2015; Cortés et al., 2015; Keller, 2010; Kohli, & Mann, 2012; Simões et al., 2012).

3.3.1. Event study framework

We follow the general guidelines by MacKinlay (1997) and Kothari and Warner (2007) for this study. The event is the announcement of the transaction, with the event window spanning from the announcement day to one year after. The estimation window for calculating abnormal returns is the year preceding the event.

In the event window, an abnormal return is the difference between the actual return obtained by the company and the normal return. The actual return is calculated using the acquirer's stock price, while the normal return is the expected performance without the event. To measure the event's impact, it is essential to estimate the normal returns and to calculate these abnormal returns.

Long-horizon tests are susceptible to the joint hypothesis problem, which combines testing whether abnormal returns are zero and whether the assumed model of expected returns is correct (Kothari & Warner, 1997). Proper risk adjustment in calculating abnormal returns is crucial, making the model for expected returns key. We use two models to estimate normal returns: the market model and the four-factor Fama-French and Carhart model.

The Fama and French (1993) three-factor model, enhanced by Carhart (1997), incorporates Size, Book-to-Market, and Momentum factors along with the market factor. These factors are crucial for isolating the incremental impact of a transaction on stock price performance, controlling from overall market influences applicable to all stocks.

For our event study on 39 transformative transactions in Spain between 2007 and 2019, the analysis will be conducted for the acquiring companies and further divided into cash and stock transactions to compare the different impacts.

3.3.2. Application of the event study

The sample of 39 Transformative Transactions, meets the conditions of homogeneity and comparability: All the acquirers are listed and have carried out a corporate transaction that has been considered transformative. All transactions have a Spanish element since either the buyer, the target, or both are Spanish companies. This would allow us to draw long-term conclusions without the contaminating effect of other relevant events that may affect the stock. Transformative Transactions are important enough so that during the year after its announcement, the acquirer's main task has been its integration.

As mentioned, with the application of an event study, our goals are twofold: to measure the long-term impact of the Transformative Transaction on the acquirer's return, and to determine if different forms of payment, cash or shares, produce different results.

8 🔄 J. REGLERO ET AL.

The market may react positively or negatively in the short term after the announcement and differently over the long term based on the transaction's impact. The form of payment is a crucial factor in structuring corporate transactions. To capture this, each observation is categorized by the form of payment: cash (using cash or debt) or stock (using newly issued shares). There are 7 transactions with a mixed payment (cash and shares) which are categorized as stock since the cash is a small percentage of the total consideration. Therefore, of the 39 observations, 26 use cash, and 13 use shares.

After categorizing the sample, the next step is to specify the parameters of the empirical study. The variable analyzed is the evolution of the acquirer's stock price, observed daily from one year before (t-1) the date of the event (t) to one year after the event (t+1).

We use the *Capital Asset Pricing Model* (Sharpe, 1964; Lintner, 1965) to calculate the expected returns. This model determines the expected return $E(R_i)$ of a stock *i* based on the risk-free rate (R_f) and its beta (β_i) relative to the market risk premium $(R_M - R_f)$. For our analysis, we use the 10-year government bond of the acquirer's country as the risk-free rate (R_f) , calculate the acquirer's beta using the previous year's returns and use the market average return from Damodaran (https://pages.stern.nyu.edu/~adamodar/), being the CAPM model for each stock *i* and time *t*:

$$E(R_{i,t}) = R_{f,t} + \beta_{i,t} * (E(R_{M,t}) - R_{f,t})....$$
[1]

Following Mackinlay (1997), returns will be indexed in event time *t*. Since we use daily observations, we define t=0 as the event day, $t = T_0$ to $t = T_1$ as the event window (length one year $L_{+1} = T_1-T_0$) and $t = T_{-1}$ to T_0 as the estimation window (length one year $L_{-1} = T_0-T_{-1}$).

We estimate beta $\hat{\beta}_{i,l-1}$ using the daily returns during the year before the event day *t*:

$$R_{i,L-1} - R_{f,L-1} = \beta_{i,L-1} * \left(R_{M,L-1} - R_{f,L-1} \right) + \varepsilon_{i,L-1} \dots$$
[2]

where $\varepsilon_{i,l-1}$ is the error term that represents the portion of the return not explained by the model, and define for each company *i* abnormal returns ($AR_{i,l+1}$) as the difference between the actual returns during the event window ($R_{i,l+1}$) and the returns predicted by the model estimation $E(R_{i,l+1} | X_{l+1})$:

$$AR_{i,L+1} = R_{i,L+1} - E(R_{i,L+1} \mid X_{L+1})$$
[3]

being X_{L+1} is the conditional information for the normal return model, in this case the event window market return. The abnormal return for each stock *i* is the disturbance term of Equation (2) calculated on an out of sample model.

As mentioned earlier, we also use the four-factor Carhart (1997) model, which expands on the three-factor Fama and French (1993) model, to estimate abnormal returns. For each stock *i* and time *t*:

$$E(R_{i,t}) = R_{f,t} + \beta_{it,1}(R_{M,t} - R_{f,t}) + \beta_{it,2} * SMB_t + \beta_{it,3} * HML_t + \beta_{it,4} * WML_t$$
[4]

where $(R_{M,t} - R_{f,t})$ is the market risk premium, SMB_t is the size factor (small minus big), HML_t is the book-to-market factor (high minus low), and WML_t is the momentum factor (winners minus losers) in time *t*. These factors control for variables affecting all companies' returns and help isolate the event's effect. To account for these factors, we used daily estimates of European factors for developed countries from the Kenneth French data library (Kenneth R. French - Data Library (dartmouth.edu).

In a similar vein as Equations (2) and (3) we calculate the abnormal returns $(AR_{i,L+1})$ using linear estimation for Equation (4).

For each model, we conduct two tests:

- 1. For the entire sample, we tested the null hypothesis (H_0) that the Transformative Transaction had no impact on the acquirer, against the alternative hypothesis (H_1) that it did have an impact.
- 2. For the two subsamples based on the method of payment, we tested the null hypothesis (H_0) that the cash and stock transactions showed similar behaviors, against the alternative hypothesis (H_1) that their behaviors were different.

4. Results

As per was explained in the previous section, we split the results of the research in (i) a descriptive analysis of the sample (ii) the event study (iii) the statistical test to confirm the robustness of the outcomes.

4.1 Descriptive analysis of the transformative transactions sample

Table 1 summarizes the performance of the quantitative variables for all acquirers of the sample of Transformative Transactions, breaking down the results for the sub-sets of transactions conducted with cash and transactions conducted with stock. Number in Table 1 are medians.

	TOTAL		CA	<u>SH</u>	<u>STOCK</u>	
	-1 YEAR	+1 YEAR	-1 YEAR	+1 YEAR	-1 YEAR	+1 YEAR
Stock price	2.5%	-5.3%	9.6%	-7.9%	-0.4%	14.4%
Dividend yield	2.4%	2.8%	2.3%	3.0%	3.0%	2.2%
Total shareholder return	7.4%	-2.6%	13.5%	-5.6%	0.2%	14.4%
Stock performance vs. Index	4.5%	-0.2%	8.1%	-1.9%	16.2%	18.2%
EV/EBITDA	12.8×	11.2×	13.0×	10.9×	12.4×	21.9×
PE	14.9×	12.9×	18.8×	12.1×	13.0×	16.7×
P/BV	2.4×	2.0×	2.4×	1.9×	2.5×	2.0×
ROE	14%	12%	14%	12%	12%	12%
Profit margin	10%	8%	9%	8%	16%	8%
Asset turnover	0.5×	0.4×	0.5×	0.4×	0.2×	0.2×
Leverage	3.3×	3.8×	3.3×	<i>4.0</i> ×	3.2×	3.5×

Table 1. Summary of the quantitative variable

The acquirer's stock market performance one year after the announcement showed a negative median of -5.3%, with cash-financed acquirers faring worse at -7.9%. In contrast, stock-financed transactions had a significant positive performance of +14.4%. This points to strong market confidence in stock-financed acquirers, as stock prices reflect both present and future expectations.

Dividends per share showed a slightly better outcome for cash transactions compared to stock transactions. Despite the debt incurred in cash transactions, many companies maintained or increased dividends to shareholders. Total shareholder return, which combines stock performance and dividends, was more favorable for stock transactions due to their positive stock market performance.

When comparing the acquirer's stock performance to the reference market index during the same period, Table 1 shows that the relative performance of stock-financed transactions was better than that of cash-financed transactions.

Regarding the valuation ratios of the acquirers, both the EV/EBITDA and PE ratios show that valuations for cash transactions declined one year after the transaction, while stock transactions saw an increase in valuation multiples. The P/BV ratio decreased for both cash and stock transactions, as the increase in book value from the target's integration was greater than the increase in the acquirer's share price.

For ROE, cash transactions showed a small decline, likely due to the positive impact of increased leverage on return on equity. This suggests that companies, on average, maintain a conservative capital structure with lower leverage than optimal.

Table 2 presents the analysis of the qualitative variables for the entire sample, and for the two sub-sets of transactions depending on the chosen form of payment.

	-		
	Total	Cash	Stock
Does the acquirer need to reduce dividends?	5	2	3
Does the acquirer need to issue additional capital?	12	6	6
Does the acquirer need to sell assets to reduce debt?	4	4	0
Does the acquirer need to change the Chairman or CEO?	4	2	2
Does the acquirer experience negative changes in the rating?	6	5	1
Does the acquirer need to undergo a restructuring process?	1	1	0

Table 2. Summary of the qualitative variables for the acquirer.

Each number shows observations with an affirmative response to the question for each criterion.

Several acquirers had to issue additional capital one year after the transaction announcement, regardless of whether the transaction was conducted with cash or stock. Additionally, some acquirers in both groups had to reduce their dividends. Six out of the 39 transactions experienced a negative change in their company's rating, with five of these being cash transactions. Four acquirers had to dispose of assets to reduce debt, and all of these were cash-financed transactions. Furthermore, one cash-financed transaction required financial restructuring, while none of the stock-financed transactions did. Additionally, there were four changes in the acquirers' Chairman or CEO, with two occurring in cash acquisitions and two in stock deals.

4.2. Event study results

With the daily share price evolution for each acquirer in the sample, in Table 3 we calculate monthly changes in both absolute terms and cumulative terms, with the acquisition announcement as the reference point.

		All acc	quirers		
	1 week	1 month	3 months	6 months	1 year
Mean	0.4%	-3.5%	-2.3%	-8.1%	-7.4%
Standard Error	0.7%	2.0%	2.8%	3.5%	5.4%
Median	-0.2%	-2.0%	-0.4%	-8.3%	-5.3%
Standard Deviation	0.05	0.13	0.17	0.22	0.34
Sample Variance	0.00	0.02	0.03	0.05	0.11
Kurtosis	0.23	3.88	-0.38	-0.45	-0.12
Skewness	0.36	-1.68	-0.53	0.25	-0.11
Range	0.21	0.62	0.64	0.89	1.42
Min	-0.08	-0.49	-0.41	-0.42	-0.83
Max	0.12	0.13	0.23	0.47	0.59
Sum	0.15	-1.35	-0.91	-3.18	-2.89
Count	39	39	39	39	39
		Cash ac	quirers		
	1 week	1 month	3 months	6 months	1 year
Mean	0.3%	-5.4%	-5.9%	-10.6%	-15.3%
Standard Error	1.0%	2.8%	3.5%	4.4%	6.1%
Median	-0.4%	-3.1%	-3.7%	-8.3%	-7.9%
Standard Deviation	0.05	0.14	0.18	0.22	0.31
Sample Variance	0.00	0.02	0.03	0.05	0.10
Kurtosis	0.19	2.66	-0.59	0.14	0.29
Skewness	0.44	-1.48	-0.38	0.42	-0.31
Range	21%	62%	63%	89%	137%
Min	-8%	-49%	-41%	-42%	-83%
Max	12%	13%	22%	47%	54%
Sum	0.07	-1.39	-1.52	-2.75	-3.97
Count	26	26	26	26	26
		Stock a	cquirers		
	1 week	1 month	3 months	6 months	1 year
Mean	0.7%	0.4%	4.7%	-3.3%	8.3%
Standard Error	1.0%	2.1%	4.0%	5.8%	9.7%
Median	-0.1%	3.7%	4.8%	2.0%	14.4%
Standard Deviation	0.04	0.07	0.14	0.21	0.35
Sample Variance	0.00	0.01	0.02	0.04	0.12
Kurtosis	-0.89	-1.61	0.38	-1.49	-0.71
Skewness	0.16	-0.45	-0.70	0.01	-0.25
Range	12%	20%	50%	59%	117%
Min	-5%	-12%	-27%	-31%	-58%
Max	7%	8%	23%	28%	59%
Sum	0.09	0.05	0.61	-0.43	1.08
Count	13	13	13	13	13

 Table 3. Descriptive statistics of the sample' share price performance evolution.

The event study results presented here are based on the analysis of one quantitative variable: the share price performance of the acquirer. Table 3 displays the basic statistics of this variable, illustrating its evolution over the year following the transaction. We examine time horizons of 1 week, 1 month, 3 months, 6 months, and 1 year. Additionally, we provide statistics for the entire sample, as well as for two subsets: cash acquirers and stock acquirers.

As far as the event study is concerned, three groups of results obtained from the event study are presented:

First, the results from the stock returns three months before versus one year after the event and the evolution during these periods, centered around the event date (Table 4, Figures 3 and 4).

	Complete Sample		S	tock	Cash		
Date	Window	Cumulative	Window	Cumulative	Window	Cumulative	
E-90	2.41%	-2.65%	4.64%	-7.85%	1.30%	-0.05%	
E-60	0.14%	-0.23%	1.68%	-3.21%	-0.63%	1.25%	
E-30	0.09%	-0.09%	1.53%	-1.53%	-0.63%	0.63%	
Event (E)	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	
E+30	-3.45%	-3.45%	0.37%	0.37%	-5.36%	-5.36%	
E+60	1.40%	-2.05%	3.84%	4.20%	0.19%	-5.17%	
E+90	-0.29%	-2.34%	0.51%	4.72%	-0.69%	-5.86%	
E+120	0.08%	-2.25%	0.38%	5.09%	-0.06%	-5.92%	
E+150	-3.45%	-5.70%	-4.89%	0.20%	-2.73%	-8.66%	
E+180	-2.44%	-8.15%	-3.48%	-3.28%	-1.93%	-10.58%	
E+210	-0.09%	-8.23%	2.91%	-0.37%	-1.58%	-12.17%	
E+240	2.70%	-5.53%	3.26%	2.89%	2.43%	-9.74%	
E+270	-0.86%	-6.39%	0.24%	3.13%	-1.41%	-11.15%	
E+300	-1.16%	-7.55%	-1.26%	1.87%	-1.11%	-12.26%	
E+330	0.71%	-6.84%	3.48%	5.34%	-0.67%	-12.93%	
E+360	-0.57%	-7.41%	2.96%	8.31%	-2.34%	-15.27%	

Table 4. Price returns per share for the total sample vs each of the subsamples according to different windows of time pre and post event.



Figure 3. Evolution of the cumulative return of the stock price for each stock in the sample from T - 90 to T+360, taking the Event as a reference point. (x axis – duration in days, y axis – share price performance).



Figure 4. Evolution of the cumulative price return for each sub-set (according to the payment method) from T - 90 to T + 360, using the event as a reference point. (x axis – time, y axis – share price performance).

12 🕢 J. REGLERO ET AL.

Second, the cumulative abnormal returns of the acquirer are compared to the market's evolution (Table 5 and Figures 5 and 6).

	Complete Sample		S	tock	Cash		
Date	Window	Cumulative	Window	Cumulative	Window	Cumulative	
E-90	2.35%	-0.92%	4.43%	-4.15%	1.31%	0.70%	
E-60	-0.29%	1.43%	-0.44%	0.28%	-0.22%	2.01%	
E-30	-1.14%	1.14%	0.16%	-0.16%	-1.79%	1.79%	
Event (E)	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	
E+30	-3.99%	-3.99%	-2.22%	-2.22%	-4.88%	-4.88%	
E+60	0.77%	-3.23%	3.09%	0.87%	-0.40%	-5.27%	
E+90	0.39%	-2.84%	1.62%	2.49%	-0.23%	-5.51%	
E+120	-0.37%	-3.21%	0.32%	2.81%	-0.71%	-6.22%	
E+150	-2.42%	-5.63%	-2.67%	0.14%	-2.30%	-8.52%	
E+180	-2.60%	-8.23%	-3.74%	-3.60%	-2.03%	-10.55%	
E+210	1.15%	-7.08%	4.98%	1.38%	-0.76%	-11.31%	
E+240	1.86%	-5.22%	3.42%	4.80%	1.08%	-10.24%	
E+270	-1.47%	-6.69%	-2.51%	2.29%	-0.94%	-11.18%	
E+300	-1.02%	-7.71%	-1.04%	1.26%	-1.01%	-12.19%	
E+330	1.19%	-6.52%	2.69%	3.95%	0.44%	-11.75%	
E+360	0.59%	-5.93%	2.32%	6.27%	-0.28%	-12.03%	

Table 5. Relative returns of price per share with respect to the market according to different windows of time from the event day.



Figure 5. Evolution of the return of the sample share price versus the reference market index of each acquirer from T-90 to T + 360. (x axis – duration in days, y axis – share price-index performance).



Figure 6. Evolution of the returns of the stocks versus the reference market index for each subsample from E - 90 to E + 360. (x axis – time, y axis – share price-index performance).

And third, the cumulative abnormal returns of the acquirer are compared to the estimated expected return as per CAPM and Fama-French-Carhart models, with the announcement of the Transformative Transaction as the event day (Tables 6 and 7).

					Expected	Stock		
					Return 1	Evolution	Difference	
		Transaction			year from	After a Year	(Abnormal	Qualitative
#	Acquiror	Date	Country	Payment Form	Event Date	from Event	Return)	Difference
1	REE	12-Feb-19	Spain	Cash	4.5%	-6.6%	-11.1%	Worse than expected
2	CIE Automotive	17-Feb-18	Spain	Cash	10.2%	-9.1%	-19.3%	Worse than expected
3	Amadeus	10-Aug-18	Spain	Cash	10.9%	-5.0%	-15.8%	Worse than expected
4	Minor Hotels	5-Jun-18	Thailand	Cash	6.9%	14.8%	7.9%	Better than expected
5	DS Smith	4-Jun-18	Germany	Cash	8.9%	-44.5%	-53.3%	Worse than expected
6	Euskaltel	16-May-17	Spain	Shares	10.2%	-19.4%	-29.6%	Worse than expected
7	Atlantia	21-Apr-17	Italy	Cash	8.8%	22.4%	13.7%	Better than expected
8	Indra	29-Nov-16	Spain	Shares	9.2%	24.5%	15.4%	Better than expected
9	Fresenius	6-Sep-16	Germany	Cash	9.5%	6.0%	-3.5%	Worse than expected
10	Endesa	27-Jul-16	Spain	Cash	5.8%	9.0%	3.2%	Better than expected
11	Merlin	21-Jun-16	Spain	Shares	12.4%	19.7%	7.3%	Better than expected
12	Gamesa	16-Jun-16	Spain	Shares	7.7%	14.4%	6.7%	Better than expected
13	Ferrovial	29-Apr-16	Spain	Cash	8.2%	3.4%	-4.7%	Worse than expected
14	BBVA	27-Jul-15	Spain	Cash	11.9%	-41.3%	-53.2%	Worse than expected
15	Merlin	9-Jun-15	Spain	Shares	10.4%	-5.3%	-15.7%	Worse than expected
16	Sabadell	19-Mar-15	Spain	Cash	9.2%	-14.7%	-23.9%	Worse than expected
17	Vidrala	14-Jan-15	Spain	Cash	6.2%	5.3%	-0.9%	Worse than expected
18	Repsol	14-Dec-14	Spain	Cash	8.6%	-36.9%	-45.5%	Worse than expected
19	Ferrovial, Macquaire	16-Oct-14	Spain	Cash	8.2%	54.1%	45.9%	Better than expected
20	Vodafone	16-Mar-14	UK	Cash	5.8%	0.2%	-5.6%	Worse than expected
21	Grifols	11-Nov-13	Spain	Cash	6.5%	-46.0%	-52.5%	Worse than expected
22	Abertis	5-Aug-12	Spain	Shares	15.5%	40.5%	25.0%	Better than expected
23	Atresmedia	14-Dec-11	Spain	Shares	13.7%	-13.6%	-27.4%	Worse than expected
24	Banco Popular	7-Oct-11	Spain	Shares	14.1%	-57.6%	-71.7%	Worse than expected
25	Schneider	31-May-11	Germany	Cash	13.9%	-24.4%	-38.3%	Worse than expected
26	ACS	1-Dec-10	Spain	Cash	14.6%	-31.6%	-46.2%	Worse than expected
27	Telefonica	28-Jul-10	Spain	Cash	10.0%	-2.6%	-12.6%	Worse than expected
28	Grifols	8-Jun-10	Spain	Shares	8.3%	36.9%	28.6%	Better than expected
29	Telecinco (Mediaset)	18-Dec-09	Spain	Shares	10.7%	-13.5%	-24.2%	Worse than expected
30	Enel	21-Feb-09	Italy	Cash	16.0%	6.1%	-9.9%	Worse than expected
31	Santander	13-Oct-08	Spain	Cash	15.0%	7.3%	-7.8%	Worse than expected
32	Gas Natural	30-Jul-08	Spain	Cash	10.0%	-60.4%	-70.3%	Worse than expected
33	Vueling	7-Jul-08	Spain	Shares	20.5%	53.5%	32.8%	Better than expected
34	Prisa	20-Dec-07	Spain	Cash	8.8%	-83.1%	-91.8%	Worse than expected
35	Iberdrola	25-Jun-07	Spain	Cash	6.8%	-21.8%	-28.7%	Worse than expected
36	Santander, RBS, Fortis	5-May-07	Spain	Shares	10.7%	59.0%	48.3%	Better than expected
37	ACS	19-Mar-07	Spain	Cash	10.1%	-23.6%	-33.7%	Worse than expected
38	BBVA	16-Feb-07	Spain	Shares	10.7%	-30.8%	-41.5%	Worse than expected
39	Colonial	20-ene-07	Spain	Cash	8.8%	-74.1%	-82.9%	Worse than expected

Table 6. Results for abnormal returns as per the CAPM model.

Table 7. Results for abnormal returns as per the Fama and French (1993) and Carhart (1997) model.

#	Acquiror	Transaction Date	Country	Payment Form	Expected Return 1 year from Event Date	Stock Evolution After a Year from Event	Difference (Abnormal Return)	Qualitative Difference
1	RFF	12-Feb-19	Snain	Cash	9.7%	-6.6%	-16.4%	Worse than expected
2	CIF Automotive	17-Feb-18	Spain	Cash	20.7%	-9.1%	-29.8%	Worse than expected
3	Amadeus	10-Aug-18	Spain	Cash	29.6%	-5.0%	-34.6%	Worse than expected
4	Minor Hotels	5-Jun-18	Thailand	Cash	-3.9%	14.8%	18.7%	Better than expected
5	DS Smith	4-Jun-18	Germany	Cash	-5.0%	-44.5%	-39.4%	Worse than expected
6	Euskaltel	16-May-17	Spain	Shares	11.2%	-19.4%	-30.7%	Worse than expected
7	Atlantia	21-Apr-17	Italy	Cash	10.7%	22.4%	11.7%	Better than expected
8	Indra	29-Nov-16	Spain	Shares	5.1%	24.5%	19.4%	Better than expected
9	Fresenius	6-Sep-16	Germany	Cash	-0.6%	6.0%	6.6%	Better than expected
10	Endesa	27-Jul-16	Spain	Cash	2.7%	9.0%	6.3%	Better than expected
11	Merlin	21-Jun-16	Spain	Shares	2.2%	19.7%	17.5%	Better than expected
12	Gamesa	16-JUn-16	Spain	Shares	-2.5%	14.4%	16.9%	Better than expected
13	Ferrovial	29-Apr-16	Spain	Cash	2.1%	3.4%	1.3%	Better than expected
14	BBVA	27-Jul-15	Spain	Cash	-25.6%	-41.3%	-15.7%	Worse than expected
15	Merlin	9-Jun-15	Spain	Shares	3.2%	-5.3%	-8.5%	Worse than expected
16	Sabadell	19-Mar-15	Spain	Cash	-29.9%	-14.7%	15.3%	Better than expected
17	Vidrala	14-Jan-15	Spain	Cash	-8.0%	5.3%	13.3%	Better than expected

Table 7. Continued.

#	Acquiror	Transaction Date	Country	Payment Form	Expected Return 1 year from Event Date	Stock Evolution After a Year from Event	Difference (Abnormal Return)	Qualitative Difference
18	Repsol	14-Dec-14	Spain	Cash	-28.2%	-36.9%	-8.8%	Worse than expected
19	Ferrovial, Macquaire	16-Oct-14	Spain	Cash	7.7%	54.1%	46.4%	Better than expected
20	Vodafone	16-Mar-14	UK	Cash	5.4%	0.2%	-5.2%	Worse than expected
21	Grifols	11-Nov-13	Spain	Cash	7.0%	-46.0%	-52.9%	Worse than expected
22	Abertis	5-Aug-12	Spain	Shares	14.6%	40.5%	25.9%	Better than expected
23	Atresmedia	14-Dec-11	Spain	Shares	18.1%	-13.6%	-31.8%	Worse than expected
24	Banco Popular	7-0ct-11	Spain	Shares	7.1%	-57.6%	-64.8%	Worse than expected
25	Schneider	31-May-11	Germany	Cash	6.6%	-24.4%	-31.0%	Worse than expected
26	ACS	1-Dec-10	Spain	Cash	-14.4%	-31.6%	-17.3%	Worse than expected
27	Telefonica	28-Jul-10	Spain	Cash	-14.6%	-2.6%	12.0%	Better than expected
28	Grifols	8-Jun-10	Spain	Shares	3.1%	36.9%	33.8%	Better than expected
29	Telecinco (Mediaset)	18-Dec-09	Spain	Shares	-4.3%	-13.5%	-9.2%	Worse than expected
30	Enel	21-Feb-09	Italy	Cash	13.9%	6.1%	-7.8%	Worse than expected
31	Santander	13-Oct-08	Spain	Cash	18.9%	7.3%	-11.7%	Worse than expected
32	Gas Natural	30-Jul-08	Spain	Cash	-2.1%	-60.4%	-58.3%	Worse than expected
33	Vueling	7-Jul-08	Spain	Shares	-29.3%	53.5%	82.6%	Better than expected
34	Prisa	20-Dec-07	Spain	Cash	-6.1%	-83.1%	-76.9%	Worse than expected
35	Iberdrola	25-Jun-07	Spain	Cash	49.0%	-21.8%	-70.8%	Worse than expected
36	Santander, RBS, Fortis	5-May-07	Spain	Shares	-6.1%	59.0%	65.1%	Better than expected
37	ACS	19-Mar-07	Spain	Cash	29.5%	-23.6%	-53.2%	Worse than expected
38	BBVA	16-Feb-07	Spain	Shares	19.0%	-30.8%	-49.9%	Worse than expected
39	Colonial	20-ene-07	Spain	Cash	26.0%	-74.1%	-100.2%	Worse than expected

Table 4 displays the average acquirer's stock return before and after the transaction, shown both in monthly terms and as accumulated returns over the analysed period. Each series represents the average evolution across the entire sample or each subset, respectively. In the table, 'Window' refers to the evolution within the specified timeframe, while 'Cumulative' indicates the evolution from each date to the event date.

The Figure 3 shows the evolution of the total sample (calculated as average evolution of the acquirers'share performance) in percentage terms from 90 days prior to the announcement of the Transformative Transaction to 360 days after the announcement. It translates numerical data of the previous table into an illustration for a simpler and more direct analysis.

During the 90 d prior to the event, the average stock price of the acquirers remained relatively flat. It is when the event occurs (announcement of the Transformative Transaction) that the return begins to decline, reaching a low point around 180 d after the event. The decline continues, remaining within a range of approximately 6%-8% for the remaining 180 d until T+360.

The payment method is a critical factor in structuring the transaction since it significantly affects the acquirer. The Figure 4 shows the average evolution of the sample in percentage terms, comparing the 90 d before the event with the 360 d after, breaking down the sample between transactions where acquirers used cash and those where they used stock.

During the 90d preceding the event, acquirers using cash for transactions experienced a sideways stock market evolution, while those using stock showed a positive performance, ranging from 7.5% to the reference value of 0 at the announcement date. In the very short term, just a few days, both groups displayed similar stock price behaviors. However, after this initial period, the stock and cash acquirers began to exhibit divergent behaviors, with these differences becoming more pronounced over time. By the end of the period, the most significant difference was observed: the average return for cash transactions was -15%, whereas stock-financed transactions showed a positive evolution of +8%.

Before conducting the event study, we compared the sample's performance (average share price evolution of acquirers) with comparable stock market indices. Table 5 presents the results, showing the returns of the acquiring companies minus the index returns in monthly intervals, both before and after the transaction. These results are provided in both monthly and cumulative terms, compared to the reference stock market where the companies are listed.

The Figure 5 shows the average evolution of the sample in percentage terms, taking the market effect into account and comparing the year before with the one after.

Figure 5 shows that from the announcement date (day 0 on the horizontal axis), the sample's performance became more unfavorable than the indexes, with a particularly negative trend in the 6 months following the announcement, which was not recovered later. Figure 6 breaks down the results by payment methods, confirming different behaviors between the sub-sets. Finally, we calculate the abnormal returns using the two models described in Section 3.3. to estimate the statistical significance of the event

Table 6 shows the one-year estimation of the AR, as explained in Equation (3).

Out of the 39 analyzed transactions, 28 had abnormal returns for the acquirer that were worse than expected one year after the transaction, while 11 transactions had better-than-expected results. The average abnormal return for the entire sample is -17.6%, with a negative median of -15.7%.

After separating the results by the form of payment, we found the following: For the 26 cash transactions, 22 had negative abnormal returns, while 4 had positive returns, with an average abnormal return of -24.7% and a median of -17.6%. For the stock transactions, 6 had negative returns, while 7 had positive returns, with an average abnormal return of -3.5% and a positive median of 6.7%.

In Table 7 we present the abnormal Returns using the Fama-French-Carhart four factor model to estimate the expected return.

In this case, out of the 39 transactions in the sample, 23 showed the acquirer's one-year returns were worse than the expected returns according to the model. In 16 transactions, the results were better. The average abnormal return for the entire sample is -11.1%, with a median of -8.8%.

When separating the results by the form of payment: For the 26 cash transactions, 17 had negative abnormal returns, while 9 had positive returns. The average abnormal return for this subsample was -19.2%, with a median of -13.7%. For the stock transactions, 6 had negative returns, while 7 had positive returns. The average abnormal return for this subsample was -5.1%, with a positive median of 16.9%.

Comparing the results obtained by both models, the CAPM model shows more negative abnormal returns overall compared to the Fama–French–Carhart model. Cash transactions generally performed worse in both models, but the Fama–French–Carhart model indicated slightly better outcomes. Stock transactions showed more positive outcomes, with the second model reflecting a higher positive median compared to the first.

4.3. Statistical analysis of the applied methodology

The last step of is to confirm the statistical significance of the event study performed, for which a statistical analysis with the Student's t-test for small samples (<100 observations) is conducted, following Mackinlay (1997). The Student's t-test is an appropriate method for small samples that follow a normal distribution, and it aims to validate the previous results and check if there are significant differences between the two data groups.

The results are included for the three assessments that compare: (1) the total sample of Transformative Transactions and each of the subsamples broken down according to the form of payment, cash or shares, (2) the two subsamples against each other, and (3) the abnormal return of the acquirer compared with the expected return as per CAPM model and Fama-French-Carhart four factor model. This test is applied to the share price evolution of the acquirer, as explained below.

As far as the first Student's t test is concerned, we compare if the stock price evolution has been different before and after the event. We propose a null hypothesis (H_0) that the sample has had an equal stock price evolution before and after the event. The alternative hypothesis (H_1) is that the stock price evolution has been unequal, and therefore, the event has had an effect causing the samples to have different behaviors. The following Table 8 shows the results obtained from applying the Student's t-test for significance.

	All Acquirers		Cash Ac	quirers	Stock Acquirersm	
	Before Event	After Event	Before Event	After Event	Before Event	After Event
Average stock Price evolution	-2.6%	-7.4%	0%	-15%	-7.8%	8.3%
Variance	0.02	0.11	0.01	0.10	0.02	0.12
Number of observations	39	39	26	26	13	13
Pearson correlation coefficient	-0.45		-0.21		-0.68	
Hypothetical difference of the means	0		-		0	
Degrees of freedom	38		25		12	
t-statistic	0.72		2.19	-	1.29	
P(T < t) one tail	0.24		0.02		0.11	
Critical value of t (one-tailed)	1.69		1.71		1.78	
$P(T \le t)$ two tails	0.47		0.04		0.22	
Critical Value of t (two-tailed)	2.02		2.06		2.18	

Table 8. Sample comparison before and after the event.

According to the Student's t-test results for the whole sample, the critical values are not significant. Therefore, it cannot be concluded that there is a difference in the stock price evolution of the sample before and after the event. Thus, the event does not have a significant impact on the acquirer's stock price evolution. The null hypothesis (H_0) is accepted. Two-tailed *p*-values of 0.47 are used for these conclusions since the sample can have both positive and negative results. However, we compare the stock price evolution before and after the event splitting the sample into to two subsets to detect differences between the subsamples.

For both sub-samples, the t-test's null hypothesis (H_0) is that stock price evolution before the event is similar to after the event. The alternative hypothesis (H_1) is that the evolution is different, indicating a significant impact from the event. For the cash acquirers' subsample, H_0 can be rejected at the 5% level with a *p*-value of .04, indicating a significantly unfavorable post-transaction stock price evolution. Before the transaction, the average return was around 0%, while after, it decreased by an average of 15%. The statistical test confirms the significance of these results. In the stock payment subsample, although there is a notable difference before and after the event (-7.8% vs. 8.3%), the statistical significance is not strong enough to confirm an impact on stock prices. The *p*-value for the entire sample is .47, while for the stock subsample, it is 0.22, indicating better results than the entire sample but not statistically significant.

As far as the second Student's t test is concerned, we compare the two subsamples and the stock price evolution after 360 d of the event is analyzed for cash and stock acquirers. Although the number of observations in each is different, 26 for cash and 13 for stock, the variances are similar, so the Student's t-test can also be applied.

The null hypothesis (H_0) proposes that the two subsamples have similar behaviors after the event, while the alternative hypothesis (H_1) suggests they differ. Behavior is measured by the stock price evo-

	Cash	Shares	
Average stock price evolution	-15.3%	8.3%	
Variance	0.10	0.12	
Number of observations	26	13	
Pearson correlation coefficient	0.1045		
Hypothetical difference of the means	0		
Degrees of freedom	37		
t-statistic	-2.15		
P(T < t) one tail	0.02		
Critical value of t (one-tailed)	1.69		
P(T<=t) two tails	0.04		
Critical Value of t (two-tailed)	2.03		

Table 9. Comparison of stock price evolution 360 d after the event between cash and stock subsamples.

lution of the acquiring companies. The following Table 9 summarizes the results obtained.

We confirmed that the two samples exhibited different behaviors after the event. Statistically, we can reject H_0 at the 5% level, with a two-tailed *p*-value of .04, indicating that the populations are distinct. These results point to stock payments being significantly more advantageous, with an 8.3% increase in performance for stock payments compared to a -15.3% decrease for cash payments.

Finally, on the third Student's t test, we compare the acquirer's abnormal stock price evolution after 360 d of the event analyzed for cash transactions and, on the other hand, for stock transactions. The null hypothesis is that the two subsamples have had similar behaviors after the event (H_0), and the alternative hypothesis is that the behaviors of the subsamples are different after the event (H_1). We measure behaviors as the abnormal stock price evolution of the acquiring companies compared with expected return as per the CAPM Model and with Fama French Carhart four factor model (Table 10).

	Abnormal vs CAPM		Abnormal vs Fama-French-Carhart	
	Cash	Stock	Cash	Stock
Average abnormal stock price evolution	-24.7%	-3.5%	-19.2%	5.1%
Variance	0.10	0.12	0.12	0.19
Number of observations	26	13	26	13
Pearson correlation coefficient	0.1052		0.1398	
Hypothetical difference of the means	0.000		0.000	
Degrees of freedom	37		37	
t-statistic	-1.916		-1.911	
P(T < t) one tail	0.032		0.032	
Critical value of t (one-tailed)	1.687		1.687	
P(T < =t) two tails	0.063		0.064	
Critical Value of t (two-tailed)	2.026		2.026	

Table 10. Comparison of the abnormal stock price evolution in the two subsamples after the event compared with expected return as per the CAPM model and Fama French Carhart model.

It is confirmed that we can reject the hypothesis that both samples had similar behaviors after the event. Statistically, we can reject H₀ at the 5% level with a one-tailed *p*-value of .03 in both tests comparing abnormal returns to CAPM and Fama-French-Carhart expected returns. Additionally, we can reject H₀ at the 10% level with a two-tailed *p*-value of .06, indicating that the populations are different.

This further confirms all previous results that stock payment is significantly more attractive in the long term for the shareholders of the acquiring company, with a difference of more than 20% in performance. The attractiveness has been measured as the stock price evolution one year after the announcement of the transaction.

To wrap up this section, we compare and summarize the findings:

- Overall sample impact. The t-test results for the entire sample indicated no significant difference in stock price evolution before and after the event. The null hypothesis (H₀) was accepted, with a two-tailed *p*-value of .47, suggesting the event did not significantly impact the acquirer's stock price.
- However, for cash payment transactions, the null hypothesis (H₀) was rejected at the 5% level, with a *p*-value of .04. This indicated a significantly unfavorable stock price evolution post-transaction, with average returns decreasing by 15% after the event.
- This was not the case for the stock payment subsample. Although there was a notable difference in stock price evolution before and after the event (-7.8% vs. 8.3%), the statistical significance for the stock payment subsample was not strong enough to confirm an impact (*p*-value of 0.22). However, stock payments showed better results than the overall sample.
- When comparing cash vs. stock acquirers, the two subsamples exhibited different behaviors after the event. H₀ was rejected at the 5% level, with a two-tailed *p*-value of .04, indicating distinct populations. Stock acquirers' were significantly more advantageous, with an 8.3% increase in performance compared to a -15.3% decrease for cash payments.
- Finally, the analysis of the abnormal returns confirmed that the two subsamples had different behaviors post-event. H₀ was rejected at the 5% level with a one-tailed *p*-value of .03, and at the 10% level with a two-tailed *p*-value of .06, in tests comparing abnormal returns to CAPM and Fama-French-Carhart expected returns. This reinforced the finding that stock payments are significantly more attractive in the long term.

5. Conclusion and discussion

This paper adds to the existing body of research about a crucial decision in corporate transactions: the choice of the payment method. Despite a wealth of literature in this domain, there are still challenges to be addressed, like those related to measuring long term performance or analyzing the effect of country specific features. In the words of Ismail and Krause (2010): 'there is still a significant gap in our understanding of the determinants of the payment form.'

In order to better estimate the long-term impact of the M&A deal on the acquirer's return, we introduced the concept of Transformative Transactions that allowed us to construct a homogeneous sample, without contaminating events, comprised of 39 comparable transactions from 2007 to 2019, in which either the buyer or the target were Spanish. The sample distinguished two sub-sets depending on the method of payment: cash (26 deals) and stock (13 deals).

We first perform a descriptive analysis on the sample, that show superior historical performance for stock transactions. We then perform an event study in which we measure the impact of the Transformative Transactions on the buyer's return, as well as the effect of the method of payment. The tests were conducted looking at the acquirer's absolute return, its relative return compared to the market's, and its abnormal return computed as per the CAPM and the Carhart (1997) four factor model which incorporates the Fama and French (1993) model. All cases confirm that Transformative Transactions have a substantial impact on the acquiring company's stock price, and that transactions with a stock consideration consistently outperform those paid in cash. Our findings have statistical significance.

These results contradict most of the existing studies analyzing the acquirer's return depending on the method of payment. Despite works like Mitchell and Stafford (2000) question these results on the grounds that merger observations are not statistically independent, most existing literature show that returns are negatively related to stock payments *vis a vis* cash payments. This effect appears to happen when looking both at short-term (Amihud et al., 1990; Asquith et al., 1983; Franks et al, 1991; Huang & Walkling, 1987; Peterson & Peterson, 1991; Savor & Lu, 2009; Travlos, 1987; Wansley et al, 1983) and long-term performance (Agrawal et al, 1992; Linn & Switzer, 1994).

Looking at this overwhelming empirical evidence, some intriguing questions arise, as stated by Eckbo et al. (1990): Why would some managers opt for stock as the payment method in acquisitions despite clear evidence predicting a negative market reaction, while others choose cash, which has no such negative effect? Are managers who choose stock unaware of or unconcerned with the empirical findings? Alternatively, are managers who select cash more knowledgeable and more focused on protecting the owners' interests?

We believe, in line with the same authors (Eckbo et al., 1990), that managers act in the owners' best interests, and try to anticipate how the market will react to their chosen payment method, weighing the benefits and costs. Therefore, managers utilize their private information to foresee market reactions and opt for the payment method expected to yield a higher abnormal return. And our test shows that the market reaction is taking into consideration several factors that can explain the choice of payment. In other words: the market is becoming savvy to different motivations behind management's decision for a certain form of payment and is not unduly penalizing stock transactions.

Among the already mentioned traditional factors influencing the choice of payment, like information asymmetry about the bidder's stock valuation, there were tax considerations. Since there is a tax disadvantage linked to cash transactions (Ayers et al., 2003, 2004), the probability of cash transactions decreases with capital gain tax. Also, several works show a link between high capital gain taxes and higher premiums paid in cash deals to compensate selling shareholders (Ayers et al., 2003; Brown & Ryngaert, 1991; Sullivan et al., 1994; Wansley et al., 1983). As Shleifer and Vishny (2003) remind us, a key ingredient in the choice for the means of payment are synergies that the market perceives in the merger. Ultimately, an M&A deal creates value if the synergistic gains more than offset the premium paid, so if cash deals entail higher premiums based on tax considerations, it could make sense that the market reacts positively to lower premium transactions, especially in jurisdictions with high capital gain tax, like Spain.

Other explanations could be related to the features of Transformative Transactions that comprise our sample. Due to the restrictive criteria applied to our definition, these transactions involve long integration periods, and probably a large relative size of the target vs. the bidder, which could impact the acquirer's capital structure in case a cash payment was chosen, jeopardizing its future growth opportunities.

Huang et al. (2012) highlight the fact that integrating two merging firms is costly, time consuming, and synergy gains will not be realized immediately. Firms expecting a longer integration period will need to reallocate more resources, delaying synergy gains. Thus, acquirers should adopt a conservative capital structure, entailing lower leverage and a larger fraction of equity, before, during, and immediately after the merger when the integration period is expected to be costly and lengthy. A long integration period is often related to the relative size of the target vs. the buyer, which will also condition the form of

payment (Draper & Paudyal, 2006; Grullon et al., 1997; Hansen, 1987; Martin, 1996; Swieringa & Schauten, 2007) and affect the leverage ratio. The financial constraints of the buyer, coupled with its growth prospects, could also affect the means of payment, making them lean towards stock instead of cash, to keep the company's financial slack (see Faccio & Masulis, 2005; Huang et al., 2012; Harford et al., 2009; Masse et al., 1990; Uysal, 2011; Vermaelen & Xu, 2014 for the effect of leverage and Boone et al., 2014; Martin, 1996; Swieringa & Schauten, 2007; Zhang et al., 2003 for how higher investment opportunities for the buyer lead to an increased use of stock financing in corporate acquisitions). Since this is the case with Transformative Transactions, we believe it justifies why the market favors stock deals.

Finally, and as pointed out by Kothari and Warner (1997), whether the apparent abnormal returns are due to mispricing, or simply the result of measurement problems is an unresolved issue among financial economists, related to the joint hypothesis issue.

We believe our empirical study on the effect of M&A consideration contributes to the literature with a novel way to filter corporate transactions so as to build a homogeneous database that will help to properly assess long term effects. It also adds to an increasing body of works that focus on local geographies with similar legal and institutional frameworks in order to separate country specific factors that affect market performance, and therefore increasing again the comparability of transactions in the sample. Finally, we show how the unexpected performance of stock deals vs cash deals is rooted in the literature and may be pointing at the market's deeper understanding of management's motivations. Our, findings, if further confirmed, have an impact on corporate managers as well as investors.

Note

1. Data from BCG (2023) M&A Report.

Authors contribution

Jesús Reglero: conceptualization, design, data curation, and analysis, writing of the first draft.

Teresa Corzo Santamaría: revising the draft and extracting the main conclusions from the analysis as well as challenging the first draft obtaining a critical revision to improve the intellectual content and academic attractiveness.

Rocío Sáenz-Diez: assisting in the initial phase of conceptualization, revising the drafts, including a critical analysis of the reviewers' comments to improve content, academic attractiveness.

All authors had contributed to the final draft. The final version of the paper has been equally approved by all authors and all authors agree to be accountable for all aspects of the work.

Disclosure of interest

No potential conflict of interest was reported by the authors.

Funding

No funding was received.

About the authors

Jesús Reglero holds a degree in Business Administration and Management from ICADE (Universidad Pontificia de Comillas, Madrid) and graduated with honors from the Master in Business Administration (MBA) at IMD (Switzerland). He is founding partner of RS Corporate Finance and Assistant Professor at Universidad Pontificia Comillas de Madrid. His research fields are corporate finance and mergers and acquisitions.

Teresa Corzo Santamaría is a Full Professor at Universidad Pontificia Comillas, ICADE, Madrid, where she teaches Derivatives and Portfolio Management at both the bachelor's and master's levels. She holds a PhD in Business and Economics and has been a CFA charterholder since 2003. At Universidad Pontificia Comillas, she served as the Head of the Finance Department at Icade Business School for four years and has held the position of Dean of the Icade

20 🕢 J. REGLERO ET AL.

School of Economics and Business Administration since 2017. Her research interests include Behavioral Finance, International Finance, and Portfolio Management.

Rocío Saénz-Diez is a Financial Management Assistant Professor at Universidad Pontificia Comillas de Madrid. Her areas of interest within Corporate Finance are Mergers and Acquisitions, Private Equity and Venture Capital activity, entrepreneurship and innovation financing. Alongside these, her research interest is wider and encompasses career management and entrepreneurial career paths and treats, as well as human resources dynamics in the workplace.

ORCID

Jesús Reglero (b) http://orcid.org/0000-0002-1114-4638 Teresa Corzo Santamaría (b) http://orcid.org/0000-0001-9535-3113 Rocío Saénz-Diez (b) http://orcid.org/0000-0002-9742-9609

Data availability statement

The data that support the construction of the Sample of Transformative Transactions, which are the base of this study are openly available in Harvard Dataverse at https://doi.org/10.7910/DVN/2WK6ON.

The data that support the findings of this study are available from the corresponding author, JR, upon reasonable request.

References

Agrawal, A., Jaffe, J. F., & Mandelker, G. N. (1992). The post-merger performance of acquiring firms: a re-examination of an anomaly. *The Journal of Finance*, 47(4), 1605–1621. https://doi.org/10.2307/2328956

Aguirreamalloa, J. (2013). La creación de valor en fusiones y adquisiciones [PhD dissertation]. Universidad de Navarra.

Akben-Selcuk, E. (2015). Do mergers and acquisitions create value for Turkish target firms? An event study analysis. *Procedia Economics and Finance*, 30, 15–21. https://doi.org/10.1016/S2212-5671(15)01250-2

- Alexandridis, G., Antypas, N., & Travlos, N. (2017). Value creation from M&As: New evidence. *Journal of Corporate Finance*, 45, 632–650. https://doi.org/10.1016/j.jcorpfin.2017.05.010
- Alexandridis, G., Petmezas, D., & Travlos, N. G. (2010). Gains from mergers and acquisitions around the world: New evidence. *Financial Management*, 39(4), 1671–1695. https://doi.org/10.1111/j.1755-053X.2010.01126.x
- Amihud, Y., Lev, B., & Travlos, N. G. (1990). Corporate control and the choice of investment financing: The case of corporate acquisitions. *The Journal of Finance*, *45*(2), 603–616. https://doi.org/10.2307/2328673

Antoniadis, I., Alexandridis, A., & Sariannidis, N. (2014). Mergers and acquisitions in the Greek banking sector: An event study of a proposal. *Procedia Economics and Finance*, 14, 13–22. https://doi.org/10.1016/S2212-5671(14)00680-7

Ashley, J. W. (1962). Stock prices and changes in earnings and dividends: some empirical results. *Journal of Political Economy*, 70(1), 82–85. https://doi.org/10.1086/258592

Asquith, P. (1983). Merger bids, uncertainty, and stockholder returns. *Journal of Financial Economics*, 11(1-4), 51–83. https://doi.org/10.1016/0304-405X(83)90005-3

Asquith, P., Bruner, R. F., & Mullins, D. W. (1983). The gains to bidding firms from merger. *Journal of Financial Economics*, 11(1-4), 121–139. https://doi.org/10.1016/0304-405X(83)90007-7

Auerbach, A. J., & Reishus, D. (1988). The effects of taxation on the merger decision in Corporate takeovers: Causes and consequences (pp. 157–190). University of Chicago Press. https://doi.org/10.7208/9780226032160-008

- Ayers, B. C., Lefanowicz, C. E., & Robinson, J. R. (2003). Shareholder taxes in acquisition premiums: The effect of capital gains taxation. *The Journal of Finance*, *58*(6), 2783–2801. https://doi.org/10.1046/j.1540-6261.2003.00622.x
- Ayers, B. C., Lefanowicz, C. E., & Robinson, J. R. (2004). The effect of shareholder-level capital gains taxes on acquisition structure. *The Accounting Review*, *79*(4), 859–887. https://doi.org/10.2308/accr.2004.79.4.859

Ball, R., & Brown, P. (1968). An empirical evaluation of accounting income numbers. *Journal of Accounting Research*, 6(2), 159–178. https://doi.org/10.2308/accr-50604

- Barber, B. M., & Lyon, J. D. (1997). Detecting long-run abnormal stock returns: The empirical power and specification of test statistics. *Journal of Financial Economics*, 43(3), 341–372. https://doi.org/10.1016/S0304-405X(96)00890-2
- Barker, C. A. (1956). Effective stock splits. Harvard Business Review, 34(1), 101–106.

BCG. (2023). M&A is looking up after bottoming out. 2023 M&A report. Boston Consulting Group.

Bharadwaj, A., & Shivdasani, A. (2003). Valuation effects of bank financing in acquisitions. *Journal of Financial Economics*, 67(1), 113–148. https://doi.org/10.1016/S0304-405X(02)00233-7

Billett, M. T., & Qian, Y. (2008). Are overconfident CEOs born or made? Evidence of self-attribution bias from frequent acquirers. *Management Science*, 54(6), 1037–1051. https://doi.org/10.2139/ssrn.687534

Blackburn, V. L., Dark, F. H., & Hanson, R. C. (1997). Mergers, method of payment and returns to manager-and owner-controlled firms. *Financial Review*, 32(3), 569–589. https://doi.org/10.1111/j.1540-6288.1997.tb00439.x

Boone, A. L., Lie, E., & Liu, Y. (2014). Time trends and determinants of the method of payment in M&As. *Journal of Corporate Finance*, *27*, 296–304. https://doi.org/10.1016/j.jcorpfin.2014.05.015

- Brown, D. T., & Ryngaert, M. D. (1991). The mode of acquisition in takeovers: Taxes and asymmetric information. *The Journal of Finance*, 46(2), 653–669. https://doi.org/10.2307/2328840
- Bruner, R. F. (1999). An analysis of value destruction and recovery in the alliance and proposed merger of Volvo and Renault. *Journal of Financial Economics*, *51*(1), 125–166. https://doi.org/10.1016/S0304-405X(98)00046-4
- Bruner, R. F. (2002). Does M&A pay? A survey of evidence for the decision-maker. *Journal of Applied Finance*, 12(1), 48–68. Bruner, R. F. (2015). *Deals from hell: M&A lessons that rise above the ashes*. John Wiley & Sons.
- Bughin, J., & Copeland, T. E. (1997). The virtuous cycle of shareholder value creation. *The McKinsey Quarterly*, 2, 156–157. Burch, T. R., Nanda, V., & Silveri, S. (2012). Taking stock or cashing in? Shareholder style preferences, premiums and
- the method of payment. Journal of Empirical Finance, 19(4), 558-582. https://doi.org/10.2139/ssrn.1569787
- Calipha, R., Tarba, S., & Brock, D. (2010). Mergers and acquisitions: a review of phases, motives, and success factors. In C. L. Cooper & S. Finkelstein (Eds.), Advances in mergers and acquisitions: Vol. 9. Advances in mergers and acquisitions (pp. 1–24). Emerald Group Publishing Limited. https://doi.org/10.1108/S1479-361X(2010)000009004
- Capron, L., & Pistre, N. (2002). When do acquirers earn abnormal returns? *Strategic Management Journal*, 23(9), 781–794. https://doi.org/10.1002/smj.262
- Carhart, M. M. (1997). On persistence in mutual fund performance. *The Journal of Finance*, 52(1), 57–82. https://doi.org/10.1111/j.1540-6261.1997.tb03808.x
- Carleton, W. T., Guilkey, D. K., Harris, R. S., & Stewart, J. F. (1983). An empirical analysis of the role of the medium of exchange in mergers. *The Journal of Finance*, *38*(3), 813–826. https://doi.org/10.2307/2328083
- Chakrabarti, R., Gupta-Mukherjee, S., & Jayaraman, N. (2009). Mars-Venus marriages: Culture and cross-border M&A. *Journal of International Business Studies*, 40(2), 216–236. https://doi.org/10.2139/ssrn.869307
- Chernenko, S., Lerner, J., & Zeng, Y. (2021). Mutual funds as venture capitalists? Evidence from unicorns. *The Review* of Financial Studies, 34(5), 2362–2410. https://doi.org/10.2139/ssrn.2897254
- Chira, I., & Madura, J. (2018). Reference points and method of payment in mergers. *Managerial Finance*, 44(2), 278–290. https://doi.org/10.1108/MF-04-2017-0109
- Cortés, L. M., García, J. J., & Agudelo, D. (2015). Effects of mergers and acquisitions on shareholder wealth: Event study for Latin American airlines. *Latin American Business Review*, *16*(3), 205–226. https://doi.org/10.1080/10978526 .2015.1075238
- Diepold, B., Feinberg, R. M., Round, D. K., & Tustin, J. (2008). Merger impacts on investor expectations: an event study for Australia. *International Journal of the Economics of Business*, 15(1), 45–62. https://doi.org/10.1080/13571510701830465
- Dilshad, M. N. (2013). Profitability analysis of mergers and acquisitions: an event study approach. Business and Economic Research, 3(1), 2781. https://doi.org/10.5296/ber.v3i1.2781
- Dolley, J. C. (1933). Characteristics and procedure of common stock split-ups. Harvard Business Review, 11(3), 316–326.
- Dong, M., Hirshleifer, D., Richardson, S., & Teoh, S. H. (2006). Does investor misvaluation drive the takeover market? *The Journal of Finance*, *61*(2), 725–762. https://doi.org/10.1111/j.1540-6261.2006.00853.x
- Draper, P., & Paudyal, K. (2006). Acquisitions: private versus public. *European Financial Management*, *12*(1), 57–80. https://doi.org/10.1111/j.1354-7798.2006.00310.x
- Duso, T., Gugler, K., & Yurtoglu, B. (2010). Is the event study methodology useful for merger analysis? A comparison of stock market and accounting data. *International Review of Law and Economics*, 30(2), 186–192. https://doi.org/10.1016/j.irle.2010.02.001
- Eckbo, B. E., & Langohr, H. (1989). Information disclosure, method of payment, and takeover premiums: Public and private tender offers in France. *Journal of Financial Economics*, 24(2), 363–403. https://doi.org/10.101 6/0304-405X(89)90052-4
- Eckbo, B. E., Giammarino, R. M., & Heinkel, R. L. (1990). Asymmetric information and the medium of exchange in takeovers: Theory and tests. *Review of Financial Studies*, 3(4), 651–675. https://doi.org/10.1093/rfs/3.4.651
- Elad, F., & Bongbee, N. (2016). Event study on the reaction of stock returns to acquisition news. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.2863243
- Emery, G. W., & Switzer, J. A. (1999). Expected market reaction and the choice of method of payment for acquistions. *Financial Management*, 28(4), 73–86. https://doi.org/10.2307/3666304
- Faccio, M., & Lang, L. H. (2002). The ultimate ownership of Western European corporations. *Journal of Financial Economics*, 65(3), 365–395. https://doi.org/10.1016/S0304-405X(02)00146-0
- Faccio, M., & Masulis, R. W. (2005). The choice of payment method in European mergers and acquisitions. *The Journal of Finance*, *60*(3), 1345–1388. https://doi.org/10.1111/j.1540-6261.2005.00764.x
- Fama, E. F., & French, K. R. (1993). Common risk factors in the returns on stocks and bonds. *Journal of Financial Economics*, 33(1), 3–56. https://doi.org/10.1016/0304-405X(93)90023-5
- Fama, E. F., Fisher, L., Jensen, M. C., & Roll, R. (1969). The adjustment of stock prices to new information. *International Economic Review*, 10(1), 1–21. https://doi.org/10.2307/2525569
- Farinós Viñas, J. E., Herrero, B., & Latorre Guillem, M. (2017). The decision to acquire listed vs. unlisted firms: determinants and value effects in the Spanish stock market. *Revista de Economía Aplicada*, 73, 55–94.
- Feito-Ruiz, I., Fernández, A. I., & Menéndez-Requejo, S. (2014). Determinants of the acquisition of listed versus unlisted firms in different legal and institutional environments. *Applied Economics*, 46(23), 2814–2832. https://doi.org/10. 1080/00036846.2014.914146

- Fernández, A. I., & Gómez-Ansón, S. (1999). Un estudio de las ofertas públicas de adquisición en el mercado de valores español. *Investigaciones Económicas*, 23(3), 471–495.
- Fishman, M. J. (1989). Preemptive bidding and the role of the medium of exchange in acquisitions. *The Journal of Finance*, 44(1), 41–57. https://doi.org/10.1111/j.1540-6261.1989.tb02403.x
- Franks, J. R., & Harris, R. S. (1989). Shareholder wealth effects of corporate takeovers: the UK experience 1955–1985. Journal of Financial Economics, 23(2), 225–249. https://doi.org/10.1016/0304-405X(89)90057-3
- Franks, J., Harris, R., & Titman, S. (1991). The postmerger share-price performance of acquiring firms. *Journal of Financial Economics*, 29(1), 81–96. https://doi.org/10.1016/0304-405X(91)90014-B
- Friedman, M. (1970). A Friedman doctrine: The social responsibility of business is to increase its profits. *The New York Times Magazine*, 13(1970), 32–33. https://doi.org/10.1007/978-3-540-70818-6_14
- Fuller, K., Netter, J., & Stegemoller, M. (2002). What do returns to acquiring firms tell us? Evidence from firms that make many acquisitions. *The Journal of Finance*, *57*(4), 1763–1793. https://doi.org/10.1111/1540-6261.00477
- García Martín, C. J., & Ferrando Bolado, M. (1992). Tomas de poder: efectos sobre la rentabilidad de las acciones. *Revista Europea de Dirección y Economía de la Empresa, 1*(1), 115–124.
- Gaspar, J., Massa, M., & Matos, P. (2005). Shareholder investment horizons and the market for corporate control. *Journal of Financial Economics*, 76(1), 135–165. https://doi.org/10.1016/j.jfineco.2004.10.002
- Gielens, K., Van de Gucht, L. M., Steenkamp, J. B. E., & Dekimpe, M. G. (2008). Dancing with a giant: The effect of Wal-Mart's entry into the United Kingdom on the performance of European retailers. *Journal of Marketing Research*, 45(5), 519–534. https://doi.org/10.1509/jmkr.45.5.519
- Golubov, A., Yawson, A., & Zhang, H. (2015). Extraordinary acquirers. *Journal of Financial Economics*, 116(2), 314–330. https://doi.org/10.2139/ssrn.2326420
- Grubb, T. M., & Lamb, R. B. (2001). Capitalize on merger chaos: Six ways to profit from your competitors' consolidation and your own. Simon and Schuster.
- Grullon, G., Michaely, R., & Swary, I. (1997). Capital adequacy, bank mergers, and the medium of payment. *Journal of Business Finance & Accounting*, 24(1), 97–124. https://doi.org/10.1111/1468-5957.00097
- Grupo IFAES. www.capitalcorporate.com (Accessed on 2023).
- Hansen, R. G. (1987). A theory for the choice of exchange medium in mergers and acquisitions. *The Journal of Business*, 60(1), 75–95. https://doi.org/10.1086/296386
- Harford, J. (2005). What drives merger waves? Journal of Financial Economics, 77(3), 529-560. https://doi.org/10.1016/j.jfineco.2004.05.004
- Harford, J., Klasa, S., & Walcott, N. (2009). Do firms have leverage targets? Evidence from acquisitions. *Journal of Financial Economics*, 93(1), 1–14. https://doi.org/10.1016/j.jfineco.2008.07.006
- Harris, M., & Raviv, A. (1988). Corporate control contests and capital structure. *Journal of Financial Economics*, 20, 55–86. https://doi.org/10.1016/0304-405X(88)90040-2
- Harris, R. S., Franks, J., & Mayer, C. (1987). *Means of payment in takeovers: Results for the UK and US*. Working paper 2456, National Bureau of Economic Research.
- Harrison, J., Oler, D., & Allen, M. R. (2005). Event studies and the importance of longer-term measures in assessing the performance outcomes of complex events. http://www.indiana.edu.
- Healy, P. M., Palepu, K. G., & Ruback, R. S. (1992). Does corporate performance improve after mergers? Journal of Financial Economics, 31(2), 135–175. https://doi.org/10.1016/0304-405X(92)90002-F
- Heron, R., & Lie, E. (2002). Operating performance and the method of payment in takeovers. *The Journal of Financial and Quantitative Analysis*, 37(1), 137–155. https://doi.org/10.2307/3594998
- Higgins, H. N. (2013). Do stock-for-stock merger acquirers manage earnings? Evidence from Japan. *Journal of Accounting and Public Policy*, 32(1), 44–70. https://doi.org/10.1016/j.jaccpubpol.2012.10.001
- Houston, J. F., & Ryngaert, M. D. (1997). Equity issuance and adverse selection: A direct test using conditional stock offers. *The Journal of Finance*, *52*(1), 197–219. https://doi.org/10.2307/2329561
- Huang, J., Pierce, J. R., & Tsyplakov, S. (2012). Post-merger integration duration and leverage dynamics of mergers: Theory and evidence. *SSRN Electronic Journal*, https://doi.org/10.2139/ssrn.1787265
- Huang, Y., & Walkling, R. A. (1987). Target abnormal returns associated with acquisition announcements: Payment, acquisition form, and managerial resistance. *Journal of Financial Economics*, 19(2), 329–349. https://doi.org/10.101 6/0304-405X(87)90008-0
- Ingham, H., Kran, I., & Lovestam, A. (1992). Mergers and profitability: a managerial success story? Journal of Management Studies, 29(2), 195–208. https://doi.org/10.1111/j.1467-6486.1992.tb00660.x
- Ismail, A., & Krause, A. (2010). Determinants of the method of payment in mergers and acquisitions. *The Quarterly Review of Economics and Finance*, *50*(4), 471–484. https://doi.org/10.1016/j.qref.2010.06.003
- Kaplan, S. N. (1989). Campeau's acquisition of Federated: Value destroyed or value added. *Journal of Financial Economics*, 25(2), 191–212. https://doi.org/10.1016/0304-405X(89)90081-0
- Karampatsas, N., Petmezas, D., & Travlos, N. G. (2014). Credit ratings and the choice of payment method in mergers and acquisitions. *Journal of Corporate Finance*, 25, 474–493. https://doi.org/10.1016/j.jcorpfin.2014.01.008
- Karolyi, G. A., Liao, R. C., & Loureiro, G. (2015). The decreasing returns of serial acquirers around the world. Cornell University. Keller, A. (2010). Competition effects of mergers: An event study of the German electricity market. *Energy Policy*, 38(9), 5264–5271. https://doi.org/10.1016/j.enpol.2010.05.034

- Kengelbach, D. J., Klemmer, D. C., Schwetzler, B., & Sperling, M. O. (2012). An anatomy of serial acquirers, M&A learning, and the role of post-merger integration, M&A Learning, and the Role of Post-Merger Integration. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.1946261
- Klasa, S., & Stegemoller, M. (2007). Takeover activity as a response to time-varying changes in investment opportunity sets: Evidence from takeover sequences. *Financial Management*, *36*(2), 1–25. https://doi.org/10.1111/j.1755-053X.2007. tb00085.x
- Kohli, R., & Mann, B. J. S. (2012). Analyzing determinants of value creation in domestic and cross border acquisitions in India. *International Business Review*, 21(6), 998–1016. https://doi.org/10.1016/j.ibusrev.2011.11.006
- Kothari, S. P., & Warner, J. B. (1997). Measuring long-horizon security price performance. *Journal of Financial Economics*, 43(3), 301–339. https://doi.org/10.1016/S0304-405X(96)00899-9
- Kothari, S. P., & Warner, J. B. (2007). Ch.1 Econometrics of event studies. In B. E. Eckbo (Ed.), Handbooks in finance: Handbook of empirical corporate finance (Vol. 1, pp. 3–36). Elsevier. https://doi.org/10.1016/B978-0-444-53265-7.50015-9
- Liargovas, P., & Repousis, S. (2011). The impact of mergers and acquisitions on the performance of the Greek banking sector: An event study approach. *International Journal of Economics and Finance*, 3(2), 89–100. https://doi.org/10.5539/ijef.v3n2p89
- Linn, S., & Switzer, J. (1994). The relation between the method-of-payment used in corporate acquisitions and subsequent firm performance. Concordia University.
- Lintner, J. (1965). Security prices, risk, and maximal gains from diversification. *The Journal of Finance*, 20(4), 587–615. https://doi.org/10.2307/2977249
- Loughran, T., & Vijh, A. M. (1997). Do long-term shareholders benefit from corporate acquisitions? *The Journal of Finance*, *52*(5), 1765–1790. https://doi.org/10.2307/2329464
- Lys, T., & Vincent, L. (1995). An analysis of value destruction in AT&T's acquisition of NCR. *Journal of Financial Economics*, 39(2-3), 353–378. https://doi.org/10.2307/2977249
- MacKinlay, A. C. (1997). Event studies in economics and finance. Journal of Economic Literature, 35(1), 13-39.
- Martin, K. J. (1996). The method of payment in corporate acquisitions, investment opportunities, and management ownership. *The Journal of Finance*, *51*(4), 1227–1246. https://doi.org/10.1111/j.1540-6261.1996.tb04068.x
- Masse, I., Kushner, J., & Hanrahan, R. (1990). The effect of the method of payment on stock returns in Canadian tender offers and merger proposals for both target and bidding firms. *Quarterly Journal of Business and Economics*, 29(4), 102–124.
- McAfee, R. P., & Williams, M. A. (1988). Can event studies detect anticompetitive mergers? *Economics Letters*, 28(2), 199–203. https://doi.org/10.1016/0165-1765(88)90114-0
- Miguel, A. D., Pindado, J., & Lozano, B. (2003). Un análisis de la creación de riqueza asociada a los fenómenos de tomas de control empresarial *Análisis Financiero Internacional*, 113, 51–65.
- Mitchell, M. L., & Stafford, E. (2000). Managerial decisions and long-term stock price performance. *The Journal of Business*, 73(3), 287–329. https://doi.org/10.1086/209645
- Moschieri, C., Ragozzino, R., & Campa, J. M. (2014). Does regional integration change the effects of country-level institutional barriers on M&A? The case of the European Union. *Management International Review*, *54*(6), 853–877. https://doi.org/10.1007/s11575-014-0206-7
- Myers, J. H., & Bakay, A. J. (1948). Influence of stock split-ups on market price. Harvard Business Review, 26(2), 251-255.
- Netter, J., Stegemoller, M., & Wintoki, M. B. (2011). Implications of data screens on merger and acquisition analysis: A large sample study of mergers and acquisitions from 1992 to 2009. *Review of Financial Studies*, 24(7), 2316–2357. https://doi.org/10.1093/rfs/hhr010
- Öberg, C., & Holtström, J. (2006). Are mergers and acquisitions contagious? *Journal of Business Research*, 59(12), 1267–1275. https://doi.org/10.1016/j.jbusres.2006.09.015
- Oler, D. K., Harrison, J. S., & Allen, M. R. (2008). The danger of misinterpreting short-window event study findings in strategic management research: An empirical illustration using horizontal acquisitions. *Strategic Organization*, 6(2), 151–184. https://doi.org/10.1177/1476127008090008
- Pardo-Garcia, C., & Sempere-Monerris, J. J. (2015). Equilibrium mergers in a composite good industry with efficiencies. SERIEs, 6(1), 101–127. https://doi.org/10.1007/s13209-014-0121-y
- Peterson, D. R., & Peterson, P. P. (1991). The medium of exchange in mergers and acquisitions. *Journal of Banking & Finance*, 15(2), 383–405. https://doi.org/10.1016/0378-4266(91)90074-V
- Pinkowitz, L., Sturgess, J., & Williamson, R. (2013). Do cash stockpiles fuel cash acquisitions? Journal of Corporate Finance, 23, 128–149. https://doi.org/10.1016/j.jcorpfin.2013.08.003
- Rani, N., Yadav, S. S., & Jain, P. K. (2015). Impact of mergers and acquisitions on shareholders' wealth in the short run: An event study approach. *Vikalpa*, 40(3), 293–312. https://doi.org/10.1177/0256090915600842
- Rau, P. R., & Vermaelen, T. (1998). Glamour, value and the post-acquisition performance of acquiring firms. *Journal of Financial Economics*, 49(2), 223–253. https://doi.org/10.1016/S0304-405X(98)00023-3
- Renneboog, L., & Vansteenkiste, C. (2019). Failure and success in mergers and acquisitions. *Journal of Corporate Finance*, *58*, 650–699. https://doi.org/10.1016/j.jcorpfin.2019.07.010
- Rhodes-Kropf, M., & Viswanathan, S. (2004). Market valuation and merger waves. *The Journal of Finance*, *59*(6), 2685–2718. https://doi.org/10.1111/j.1540-6261.2004.00713.x

- Ruback, R. S. (1983). Assessing competition in the market for corporate acquisitions. *Journal of Financial Economics*, 11(1–4), 141–153. https://doi.org/10.1016/0304-405X(83)90008-9
- Sankar, B. P. B., & Leepsa, N. M. (2018). Payment methods in mergers and acquisitions: A theoretical framework. International Journal of Accounting and Financial Reporting, 8(1), 170. https://doi.org/10.5296/ijafr.v8i1.12354
- Savor, P. G., & Lu, Q. (2009). Do stock mergers create value for acquirers? *The Journal of Finance*, 64(3), 1061–1097. https://doi.org/10.1111/j.1540-6261.2009.01459.x
- Schwert, G. W. (2000). Hostility in takeovers: in the eyes of the beholder? The Journal of Finance, 55(6), 2599–2640. https://doi.org/10.1111/0022-1082.00301
- Shah, P., & Arora, P. (2014). M&A announcements and their effect on return to shareholders: An event study. Accounting and Finance Research, 3(2), 170–190. https://doi.org/10.5430/afr.v3n2p170
- Sharpe, W. F. (1964). Capital asset prices: A theory of market equilibrium under conditions of risk. *The Journal of Finance*, *19*(3), 425–442. https://doi.org/10.1111/j.1540-6261.1964.tb02865.x
- Shleifer, A., & Vishny, R. W. (2003). Stock market driven acquisitions. *Journal of Financial Economics*, 70(3), 295–311. https://doi.org/10.1016/S0304-405X(03)00211-3
- Simões, M. D., Macedo-Soares, T. D. L., Klotzle, M. C., & Pinto, A. C. F. (2012). Assessment of market efficiency in Argentina, Brazil and Chile: An event study of mergers and acquisitions. *Brazilian Administration Review*, 9(2), 229–245. https:// doi.org/10.1590/S1807-76922012000200007
- Straub, T. (2007). Reasons for frequent failure in mergers and acquisitions: A comprehensive analysis. Springer Science & Business Media. https://doi.org/10.1007/978-3-8350-9637-0
- Stulz, R. (1988). Managerial control of voting rights: Financing policies and the market for corporate control. *Journal of Financial Economics*, 20, 25–54. https://doi.org/10.1016/0304-405X(88)90039-6
- Sullivan, M. J., Jensen, M. R., & Hudson, C. D. (1994). The role of medium of exchange in merger offers: examination of terminated merger proposals. *Financial Management*, 23(3), 51–62. https://doi.org/10.2307/3665621
- Swieringa, J., & Schauten, M. (2007). The payment method choice in Dutch mergers and acquisitions. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.1018899
- Tanna, S., Yousef, I., & Nnadi, M. (2020). Probability of mergers and acquisitions deal failure. Journal of Financial Economic Policy, 13(1), 1–30. https://doi.org/10.1108/JFEP-09-2019-0182
- Travlos, N. G. (1987). Corporate takeover bids, methods of payment, and bidding firms' stock returns. *The Journal of Finance*, 42(4), 943–963. https://doi.org/10.1111/j.1540-6261.1987.tb03921.x
- Uhlenbruck, K., Hitt, M. A., & Semadeni, M. (2006). Market value effects of acquisitions involving Internet firms: A resource-based analysis. *Strategic Management Journal*, *27*(10), 899–913. https://doi.org/10.1002/smj.546
- Uysal, V. B. (2011). Deviation from the target capital structure and acquisition choices. *Journal of Financial Economics*, 102(3), 602–620. https://doi.org/10.1016/j.jfineco.2010.11.007
- Vermaelen, T., & Xu, M. (2014). Acquisition finance and market timing. Journal of Corporate Finance, 25, 73–91. https:// doi.org/10.1016/j.jcorpfin.2013.11.004
- Wansley, J. W., Lane, W. R., & Yang, H. C. (1983). Abnormal returns to acquired firms by type of acquisition and method of payment. *Financial Management*, 12(3), 16–22. https://doi.org/10.2307/3665512
- Yoo, K., Lee, Y., & Heo, E. (2013). Economic effects by merger and acquisition types in the renewable energy sector: An event study approach. *Renewable and Sustainable Energy Reviews*, 26, 694–701. https://doi.org/10.1016/ j.rser.2013.06.002
- Yook, K. C. (2003). Larger return to cash acquisitions: Signaling effect or leverage effect? *The Journal of Business*, *76*(3), 477–498. https://doi.org/10.1086/375255
- Yook, K. C., Gangopadhyay, P., & McCabe, G. M. (1999). Information asymmetry, management control, and method of payment in acquisitions. *Journal of Financial Research*, 22(4), 413–427. https://doi.org/10.1111/j.1475-6803.1999. tb00703.x
- Zaheer, A., Hernandez, E., & Banerjee, S. (2010). Prior alliances with targets and acquisition performance in knowledge-intensive industries. *Organization Science*, *21*(5), 1072–1091. https://doi.org/10.1287/orsc.1100.0528
- Zhang, P., Wang, P., Jones, T., & Region, P. (2003). What Determine Payment Methods in Mergers and Acquisitions? Manchester School of Management, University of Manchester. (Working Paper No.0103). https://doi.org/10.1016/j. jhazmat.2021.127959
- Zollo, M., & Meier, D. (2008). What is M&A performance? Academy of Management Perspectives, 22(3), 55–77. https:// doi.org/10.5465/amp.2008.34587995

Appendix A

Year	Acquiror	Target	Sector Target
2019	REE	Hispasat	TMT
2018	CIE Automotive	Inteva Roof Systems	Industrial
2018	Amadeus	TravelClick	TMT
2018	Minor Hotels	NH Hoteles	Tourism
2018	DS Smith	Europac	Industrial
2017	Euskaltel	Telecable	TMT
2017	Atlantia	Abertis	Infrastructure
2016	Indra	Tecnocom	TMT
2016	Fresenius	Quironsalud	Health
2016	Endesa	Enel Green Power	Energy
2016	Merlin	Metrovacesa	Real Estate
2016	Gamesa	Siemens Wind Power	Energy
2016	Ferrovial	Broadspectrum	Infrastructure
2015	BBVA	Garanti	Banking
2015	Merlin	Testa	Real Estate
2015	Sabadell	TSB	Banking
2015	Vidrala	Encirc	Industrial
2014	Repsol	Talismann	Energy
2014	Ferrovial, Macquarie	NDH1	Infrastructure
2014	Vodafone	Ono	TMT
2013	Grifols	Novartis (filial)	Pharmaceuticals
2012	Abertis	OHL Brasil	Infrastructure
2011	Atresmedia	La Sexta	TMT
2011	Banco Popular	Banco Pastor	Banking
2011	Schneider	Telvent	TMT
2010	ACS	Iberdrola	Energy
2010	Telefonica	Vivo	TMT
2010	Grifols	Talecris	Pharmaceuticals
2009	Telecinco (Mediaset)	Cuatro + 22% Digital Plus	TMT
2009	Enel	Endesa	Energy
2008	Santander	Sovereign Bancorp	Banking
2008	Gas Natural (Naturgy)	Union Fenosa	Energy
2008	Vueling	Clickair	Tourism
2007	Prisa	Sogecable	TMT
2007	Iberdrola	Energy East	Energy
2007	Santander, RBS, Fortis	ABN Amro	Banking
2007	ACS	Hochtieff	Construction
2007	BBVA	Compass	Banking
2007	Colonial	Biofisa	Pool Ectato

Table A1. Sample of transformative transactions.